Author’s response to reviews

Title: Robotic versus laparoscopic distal pancreatectomy: an up-to-date meta-analysis

Authors:

Gian Piero Guerrini (guerrinigp@yahoo.it)
Andrea Lauretta (andrea.lauretta@yahoo.it)
Claudio Belluco (cbelluco@cro.it)
Matteo Olivieri (molivieri@cro.it)
Marco Forlin (mforlin@cro.it)
Stefania Basso (mbasso@cro.it)
Bruno Breda (bbreda@cro.it)
Giulio Bertola (gbertola@cro.it)
Fabrizio Di Benedetto (fdiben@unimore.it)

Version: 2 Date: 17 Sep 2017

Author’s response to reviews:

Dear Hayley Henderson

Editors in chief
BMC Surgery

Thank you for the opportunity of reviewing this manuscript.

I have made all the text changes as you requested me. Thank you for accepting the article for publication in your journal.

Yours Faithfully,

Gian Piero Guerrini Md, PhD

1. Please change the Introduction heading to Background.
2. Please add a “Conclusions” section after the “Discussion” section. This should state clearly the main conclusions of the research article and give a clear explanation of their importance and relevance.

Author: I have made this change to the text

3. Please provide figure titles/legends under a separate heading of 'Figure Legends' after the References. If Figure titles/legends are within the main text of the manuscript, please move them.

Author: I have added the Figure legend to the text

Figure files should contain only the image/graphic, as well as any associated keys/annotations. If titles/legends are present within the figure files, please remove them.

4. Figures should be provided as separate files, and each figure of a manuscript should be submitted as a single file.

Author: I have made this change to the text

5. Please represent authors' names using their full initials, not their full name, in the Authors' Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

I have ve put the initials of the authors in the Authors' Contributions section

6. We have noted that the contributions for Olivieri Matteo have been indicated using the name Olivieri, not Matteo. Please check that this author's name is the right way round on both the title page and in Editorial Manager, as it appears that his last name has been put down as his first name.

Author: I Have corrected the name of this author whose family name is Olivieri

7. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
Author: I have followed this procedure.